Author's response to reviews

Title: Complementary and Alternative Medicine Use among Breast Cancer Survivors: The Black Women's Health Study

Authors:

Mireille Bright-Gbebry (mjg222@georgetown.edu)
Kepher H Makambi (khm33@georgetown.edu)
JoyAnn Phillips Rohan (jnp28@georgetown.edu)
Lynn Rosenberg (lrosenbe@bu.edu)
Julie R Palmer (jpalmer@bu.edu)
Adana A Llanos (a1326@georgetown.edu)
Lucile L Adams-Campbell (lla9@georgetown.edu)

Version: 3 Date: 11 February 2011

Author's response to reviews: see over
February 11, 2011

Dear Sir/Madam:

Re: Manuscript entitled, "Use of Multivitamins, Folic Acid and Herbal Supplements among Breast Cancer Survivors: The Black Women’s Health Study" # 00356564

Please accept the revisions of the above cited manuscript. We have further improved the manuscript based on the following suggestions:

- **Reviewer 2**
  
  Comment 1: Move sentence about years since diagnosis to the end of paragraph.
  
  Response 1: We have moved the sentence to the end of paragraph 1 of the results section. (See results, pg 7)

- **Format changes**
  
  - Abstract: We have included current knowledge related to CAM usage as suggested and have updated the abstract details on the submission page.
  - Section headings: We have removed all unnecessary capital letters in all section headings.
  - Competing interests: We have revised accordingly.
  - References: The reference list is included in the revised manuscript.
  - Highlighting/Tracking: We have removed the yellow highlighting and other track changes from the manuscript.

Thank you for your consideration of this manuscript.

Sincerely,

Lucille L. Adams-Campbell, PhD
Professor of Oncology
Associate Dean for Community Health & Outreach
Associate Director, Minority Health & Health Disparities Research